Math 103:99

Student information

Name [PLEASE PRINT!]

E-mail address [PLEASE PRINT!]

(Possible) Major(s)

I would like to distribute the information above to students in the class and also post it on a web page connected with this course. I hope that this will help students get in touch with each other and work together.

Do you consent to this? Please circle one of these:

YES NO

Briefly, why are you taking this course?

Your first homework assignment

You should get e-mail from me within two days. Please answer it before the next class. You’ll need to inspect the web to respond.